Author's response to reviews

Title: Congenital syphilis, still a reality in 21st century: a case report

Authors:

Monica Chaudhary (drmonica74@gmail.com)
Bineeta Kashyap (dr_bineetakashyap@yahoo.co.in)
Preena Bhalla (preenadr@gmail.com)

Version: 5 Date: 30 August 2007

Author's response to reviews: see over
To,
The editor,
JMCR

Sub: Cover letter for final formatting changes in the manuscript

Sir,

Thanks for accepting our manuscript for publication in your esteemed journal. We have made the required formatting changes to the manuscript and below are the changes made:

1. Title changed as asked.
2. Authors listed as asked and all qualifications removed. The institutional affiliations are marked as superscript numbers and corresponding author identified by *.
3. Affiliation has been written in full and e-mail ID’s of rest of the authors given below the details of corresponding author.
4. All headings and subheadings are given in sentence case; underlining has been removed and aligned left.
5. Introduction changed to background and case report changed to case presentation
6. References have been cited in the text in square brackets and where ever there were more than 3 names, have been replaced by et al.

I hope the changes are satisfactory and all care has been taken to ensure compliance to all asked changes.

Thanking you.

Dr. Monica Chaudhary
Asstt. Professor,
Deptt. of Microbiology
Maulana azad Medical College,
New Delhi-02